**Author’s response to reviews**

**Title:** Evaluation and refinement of the PRESTARt tool for identifying 12-14 year olds at high lifetime risk of developing type 2 diabetes compared to a clinicians assessment of risk: a cross-sectional study

**Authors:**

Laura Gray (lg48@le.ac.uk)

Emer Brady (emer.brady@uhl-tr.nhs.uk)

Olatz Albania (oalbaina@kronikgune.org)

Charlotte Edwardson (ce95@leicester.ac.uk)

Deirdre Harrington (dh204@leicester.ac.uk)

Kamlesh Khunti (kk22@le.ac.uk)

Joanne Miksza (jkm42@leicester.ac.uk)

Joao Raposo (filipe.raposo@sapo.pt)

Ellesha Smith (eas24@leicester.ac.uk)

Andriani Vazeou (agerasim@gmail.com)

Itziar Vergara (itziar.vergaramitxeltorena@osakidetza.eus)

Susann Weihrauch-Blüher (Susann.Blueher@medizin.uni-leipzig.de)

Melanie Davies (melanie.davies@uhl-tr.nhs.uk)

**Version:** 6 **Date:** 12 Jul 2019

**Author’s response to reviews:**

We have clarified that written informed consent was provided by parents. We have added the case report form and questionnaire used to collect data from participants as additional files.